Author's response to reviews

Title: Renal effects of dexmedetomidine during coronary artery bypass surgery: a randomized placebo-controlled study

Authors:

Kari Leino (kari.leino@tyks.fi)
Markku Hynynen (markku.hynynen@hus.fi)
Jouko Jalonen (jouko.jalonen@utu.fi)
Markku Salmenperä (markku.salmenpera@hus.fi)
Harry Scheinin (harry.scheinin@utu.fi)
Riku Aantaa (riku.aantaa@tyks.fi)
Dexmedetomidine in Cardiac Surgery Study Group (DCSS Group) (anne.kuitunen@kolumbus.fi)

Version: 4 Date: 26 April 2011

Author's response to reviews:

Dear BMC Editorial Production Team,

Thank you for the acceptance of our manuscript. We have made the required major and minor revisions with the exception that the figures were left untouched. We were not confident should we report the email addresses of the authors included in the study group and how should we express the affiliations at the end of the manuscript after the required changes. We rely on your expertise and fully accept further editing these or possibly other items may require by you.

Sincerely yours,

Kari Leino